

The Relationship between Dividend Changes and Future Earnings Changes

Master Thesis Finance

Written by: Yilin Li
ANR: 243331
Date: July, 2014

Supervisor: Mintra Dwarkasing

Master Thesis Finance by Yilin Li*
Supervised by Mintra Dwarkasing
Tilburg University, July 2014

^{*} I would like to express my gratitude to Mintra Dwarkasing who has offered me valuable suggestions in the academic study. Without her professional guidance and patient instruction, the completion of this thesis could not have been possible.

Table of Content

Abstract	4
1. Introduction	5
1.1 The Context of the Research	5
1.2 Structure of the Research	7
2. Literature Review	8
2.1 Trends in Dividends: Payers and Payout	8
2.1.1 Disappearing Dividends and the Evidence	
2.1.2 Possible Explanations for the Disappearing Dividends	
2.2 Agency Costs and the Free Cash Flow Hypothesis	10
2.3 Information Asymmetry and Dividend Signaling Theory	12
2.3.1 Dividend Signaling Models	12
2.3.2 Empirical Evidence on Dividend Signaling	13
2.4 Firm Life Cycle Theory of Dividend	16
3. Data & Methodology	18
3.1 Hypothesis Development	18
3.1.1 Hypothesis 1: Dividend Changes are Informative about Future Earnings	
3.1.2 Hypothesis 2: Dividend Changes are Informative about Future Earning,	
Controlling for Past Earnings Changes and the Lagged ROE	18
3.1.3 Hypothesis 3: Dividend Changes are Positively (Negatively) Related to Futu	ıre
Earnings Changes for the Corresponding (Contradictory) Sign Group	19
3.2 Variable Construction	19
3.3 Dataset	20
3.4 Methodology & Empirical Results	24
3.4.1 Model Specifications	24
3.4.2 Empirical Tests of the Models	25
4. Robustness Checks	32
5. Summary and Recommendations	35
5.1 Summary of the Research	
5. 2 Recommendations	
Deferences	20

Abstract

This study examines the relationship between dividend changes and future earnings changes for industrial firms listed in NYSE, AMEX, and NASDAQ from 1993 to 2011. This study is based on the model developed by Nissim and Ziv (2001) who find a positive relation between dividend changes and earnings changes in two years following the dividend change. Contrary, I find no evidence supporting that dividend changes are informative about future earnings changes based on the whole sample data. Both dividend increases and dividend decreases are unrelated to earnings changes in the two subsequent years. Beyond Nissim and Ziv (2001), I additionally differentiate dividend changes between confirmatory dividend changes (observations that the dividend change and recent past earnings change are of the same sign) and contradictory dividend changes (observations that the dividend change and recent past earnings change have opposite signs) in the same way as Koch and Sun (2004) who examine the market reactions to dividend change announcements by distinguishing dividend changes between confirmatory dividend changes and contradictory dividend changes. I find that confirmatory dividend changes (i.e. the dividend change and past earnings change are of the same sign)) are positively related to earnings changes in each of the two subsequent years, and contradictory dividend decreases (i.e. dividends decrease when past earnings went up.) are negatively related to earnings changes in the second year following the dividend change. However, these empirical results are not robust to my alternative measurement of the deflator. Additionally, the mean and median cumulative abnormal stock returns for the 2-day period surrounding the announcement of the dividend change are positive for dividend increases, and negative for dividend decrease. Overall, this study finds that dividend changes provide substantially limited information about changes in the firm's future earnings.

Master Thesis Finance 2014

1. Introduction

Dividend policy refers to the payout policy that a firm follows in determining the size and pattern of cash distributions to shareholders over time. A company's board of directors, with the input of senior management, sets a corporation's dividend policy. Miller and Modigliani (1961) propose the financial irrelevance theory in their pioneering work, stating that dividend policy has no impact on overall firm value. They document that investors do not care about receiving capital gains or dividends because the total returns will be the same. But the model is developed in an idealized world with perfect capital market, rational investor behavior, and perfect certainty. In the real world, the model is not realistic. If we relax one or more assumptions of the dividend irrelevance hypothesis, it is found that dividend policy does seem to matter for firm value. Most previous studies (e.g. Pettit (1972, Brickley (1983) and Lie (2000)) present similar results showing that the stock market reacts positively to dividend increases and initiations but negatively to dividend decreases and omissions. In other words, dividend policy does affect firm value.

In practice, determining an appropriate dividend policy is often difficult because the firm has to balance many potentially conflicting factors. Managers need to consider their long-term financing goals as well as shareholder value maximization. Therefore, they cannot decide dividend policy isolated from other interrelated decisions as dividend policy affects the amount of the firm's retained earnings that are important to exploit future growth as well as shareholder wealth.

1.1 The Context of the Research

Lintner (1956) develops a theoretical dividend model which gives us implications about the corporate dividend decision making. His model suggests that firms tend to set a long-term target payout ratio and adjust it slowly over time. Firms tend to increase dividends only when managers believe that cash flows in the future would be sufficient to support the higher dividend rate. And firms tend to avoid dividend cuts if at all possible. So dividend has a smoother pattern than earnings because managers try to stabilize dividend payments.

Based on Lintner's (1956) finding, the dividend signaling model has emerged to explain why firms pay dividends and occasionally adjust dividend payments. The

¹ Baker, H.K. (2009), *Dividends and Dividend Policy*. JohnWiley & Sons, Inc., Hoboken, New Jersey, pp. 3

dividend signaling model suggests that dividend changes provide information content about future profitability. Due to the information asymmetry between managers and outside investors, managers use the dividend change as a signaling device to convey their expectations about the firm's future profits.

Empirical studies have long investigated the relationship between dividend change and future profitability. Healy and Palepu (1988) find that dividend initiations are associated with positive earnings in the year before, and two years after the dividend initiation. Conversely, dividend omission is associated with earnings declines in the two years before and one year after the dividend omission. More recently, Nissim and Ziv (2001) find supporting evidence of the information content of dividends. They find that dividend changes are positively related to earnings changes in each of the two years subsequent to the dividend change.

My study is an extension of the study done by Nissim and Ziv (2001). First following Nissim and Ziv (2001), I will study how dividend changes are related to earnings changes in each of the two years following the dividend change based on industrial firms listed in NYSE, AMEX, and NASDAQ from 1993 to 2011. Also, motivated by Koch and Sun (2004) who examine the market reactions to dividend change announcements by distinguishing dividend changes between confirmatory dividend changes (observations for which the dividend change and recent past earnings change are of the same sign) and contradictory dividend changes (observations for which the dividend change and recent past earnings change have opposite signs), I will additionally divide my firm-year observations that have made non-zero dividend changes into two subgroups in the same way as Koch and Sun (2004). I hypothesize that confirmatory dividend changes are likely to be positively related to earnings changes in the two subsequent years. One of the possible explanations is that dividends are paid out with current or prior earnings. In the most case, change in dividends follows change in past earnings in the same direction. So these confirmatory dividend changes are likely to signal that the direction of past earnings change is expected to be sustainable in the future. Conversely, I hypothesize that contradictory dividend changes are likely to be negatively related to earnings changes in the two subsequent years.² One of the possible explanations for the negative relation is that for firms with contradictory dividend decreases, managers may have seen the potential investment

-

² Of course, I am aware that contradictory dividend changes may be positively related to earnings changes in the two subsequent years. For instance, the manager may want to keep more funds available for his own sake when the past earnings increase is followed by the dividend cut. Also, the manager may be confident about his company and believe that the firm's future performance will get better to continuously support the increased dividends.

opportunities and intend to keep more funds for future investments which are likely to increase subsequent earnings, and that for firms with contradictory dividend increases, managers may want to satisfy equity investors through increasing dividends but less funds will be available for future development since they are paying more dividends in the context of earnings decreases, then their future earnings may get worse. Although there may be other reasons for these firms doing so, and different reasons may lead to different outcomes. In my study, I only hypothesize that contradictory dividend changes are likely to be negatively related to earnings changes in the two subsequent years and test whether my hypothesis can be supported by the empirical results.

Moreover, many researches (e.g. Pettit (1972), Kaly and Loewenstein (1985)) have documented the information content of dividends by examining the relationship between dividend changes and abnormal stock returns around the dividend change announcement. Most of these studies find that dividend increases are associated with positive abnormal stock returns in the days surrounding the dividend change announcement, and dividend decreases are associated with negative abnormal stock returns in the days surrounding the dividend change announcement. In the descriptive statistics section, I will display the mean and median cumulative abnormal stock returns for the 2-day period surrounding the announcement of the dividend increase and decrease, respectively.

My study contributes to current literature in several ways: firstly, beyond Nissim and Ziv (2001), I will examine whether dividend changes play a different role in confirmatory dividend changes group and contradictory dividend changes group. As far as I know, it has not been investigated before. Secondly, my sample consists of U.S. publicly traded firms during the recent twenty years from 1993 to 2011. Most previous studies employ the data in the period before the early 1990s, their findings may be less indicative for current and future investors. Hopefully, my research may fill the gap in the existing literature and provide some new information to the investors and firm managers who make the dividend policy in corporate finance.

1.2 Structure of the Research

The structure of the research is organized as follows. Section 2 presents the literature review covering various dividend models and relevant empirical studies during the past few decades. Section 3 provides hypothesis development, sample selection criteria, variable construction, empirical methods, empirical results and analysis. Section 4 provides robustness checks of my empirical results. Finally, Section 5 contains the

summary and conclusions.

2. Literature Review

This section provides trends in dividend payments over time and various dividend theories that offer some implications about why firms choose to pay dividends and the role dividends play in corporate finance.

2.1 Trends in Dividends: Payers and Payout

This section presents the disappearing pattern of the fraction of publicly traded firms paying dividends and provides some possible explanations for the disappearing incidence.

2.1.1 Disappearing Dividends and the Evidence

Researchers have long debated on what makes most public firms to pay out cash in the form of dividend. Miller and Modigliani (1961) propose the dividend irrelevance proposition, stating that firm value is not affected by its dividend policy, investors are indifferent between receiving capital gains and receiving cash dividend as the total returns from price appreciation and dividend will be same. Easterbook (1984) and Jensen (1986) propose the proposition to explain the payout to shareholders based on the agency cost. They argue that distributing cash to shareholders will reduce the amount of funds under the manager's control, thus payout can alleviate the overinvestment problem by self-served managers who might use the excess cash to invest in value-reducing projects.

In the United States, tax on dividend is higher than that on capital gains. Theoretically, companies should prefer share repurchase rather than dividends from the tax perspective. However, in the real world, the majority of listed firms still prefer to payout cash to shareholders. Black (1976) proposes the *dividend puzzle* stating that firms generally pay dividends to shareholders despite the tax disadvantage of dividends.

Recently, Fama and French (2001) find a decline incidence in the percentage of dividend paying firms. They document that the propensity of firms paying cash dividends has decreased dramatically after 1978. After reaching a peak of 66.5 percent in 1978, the proportion of firms paying cash dividends declines continuously to 20.8 percent in 1999. On average, about 5 percent of sample firms stop paying dividends. Despite the surge in the number of public firms, the number of payer has decreased

sharply. Baker and Wurgler (2004a) find the similar dividend-paying pattern in line with Fama and French (2001). They assert that the declining trend in propensity of firms paying dividends is not a new phenomenon.

2.1.2 Possible Explanations for the Disappearing Dividends

Fama and French (2001) identify three important factors that determine the decision to pay dividends: profitability, investment opportunities and firm size. They attribute the declining incidence of firms paying dividends to the firm characteristics of the newly listed firms after 1978. These newly listed firms after 1978 tilt towards characteristics of small size but fast-growing, both of which are strongly associated with low probability to pay dividends. They further point out firms are less likely to pay dividends, even among large and profitable firms. DeAngelo, DeAngelo and Skinner (2000) provide supplemental evidence showing that special dividends have also disappeared recently. Firms used to pay special dividends almost as predictably as profitable companies pay regular cash dividends.

Grullon and Michaely (2002) argue that share repurchases have emerged as an economically significant alternative to regular cash dividends in the early 1980s. The aggregate expenditure on the repurchase of common and preferred shares have more than tripled from \$9.2 billion to \$28.6 billion between 1983 and 1984. Grullon and Michaely (2002) also argue that firms finance their share repurchase with funds that they otherwise would have paid out as cash dividends. The growing share repurchases during the 1980s and 1990s matches almost perfectly with the declining cash dividend payouts. Using Lintner's (1956) dividend model, they find evidence supporting their proposition that firms seem to have substituted share repurchase for regular cash dividends, even after controlling for the cross-sectional variation in firm characteristics. They point out that before 1982, managers were uncertain about the legitimacy of share repurchases since there were no legal guidelines for repurchases at that time. Hence, few managers would take the risk of being charged with market manipulation. The Securities and Exchange Commission (SEC) established guidelines for repurchases on the open market in 1982, which provide managers with a safe harbor to carry on repurchasing shares.

On the other hand, share repurchases do not constitute a long-term commitment as the regular cash dividends, Lintner's (1956) dividend model suggest that dividends are an ongoing commitment to distribute cash to shareholders. Jagannathan, Stephens and Weisbach (2000) document that share purchases are mainly used to distribute temporary cash flows. As a consequence, share repurchases are very volatile because

these firms typically have more uncertain cash flows. Graham, Harvey, and Michaely (2005) provide further evidence showing that managers tend to use share repurchases to pay out transitory cash flows. They document that many managers now favor share repurchases because share repurchases are supposed to be more flexible than regular cash dividends. More interestingly, they find that when choosing between share repurchases and regular dividends, most managers do not think that tax consideration is a dominant factor.

Amihud and Li (2006) provide another explanation for the disappearing dividends based on the dividend signaling. They suggest that dividends are disappearing because of the decline in the information content of dividends. Paying dividends is costly because the firm might have to raise new funds once it pays out these funds. With the information conveyed in dividends decreasing, managers are becoming unwilling to use dividend as a signaling device. They attribute the declining information content of dividends to the shareholder structure changes. They find that the average portion of a firm's equity held by institutional investors has increased from 29 percent to 53 percent between 1980 and 1990. Generally, institutional investors have the resources and power to gather and process information which retail investors are unable to do. They are better informed in this way rather than through dividends. They also argue that stock price has already incorporated part of the information that is supposed to be conveyed by the dividend announcement.

To conclude, there is no single explanation for the disappearing incidence of dividends after 1978.

2.2 Agency Costs and the Free Cash Flow Hypothesis

The free cash flow hypothesis traces its root to the agency problem between firm insiders and outside investors as has been documented by Jensen and Meckling (1976). An obvious characteristic of public firms is the divergence of interest between managers and shareholders. Jensen (1986) argues that managers have incentives to expand the firm bigger than its optimal size since more resources will be under their control as the company becomes larger. This selfish motivation will induce managers to invest in value-reducing projects or even negative net present value (NPV) projects, which might benefit themselves but undermine shareholders' interest. Since managers are less likely to use external funds to finance the value-reducing or negative NPV investments, they rely on internally generated funds. Hence, firms with substantial amounts of free cash flows are likely to overinvest in unnecessary projects.

Easterbrook (1984) and Jensen (1986) suggest that to alleviate the agency problem arising from the separation of ownership and management, one possible solution is to return excess funds to shareholders by paying cash dividends. Many studies have documented that managers are reluctant to cut dividend payments (e.g. Lintner (1956)), so a dividend initiation or increase means reducing excess cash at the disposal of managers, thereby reducing their ability to misuse these funds. ³ The implication from the free cash flow hypothesis is that dividend increase announcements should be followed by positive stock market reactions because investors are aware of the agency problem ex-ante and believe that the potential overinvestment problem is mitigated by dividend increases. On contrary, dividend decrease should be accompanied by negative market reactions because they are not a preferable signal to investors.

Lang and Litzenberger (1989) are the first to perform empirical test on the free cash flow hypothesis. They hypothesize that market return to dividend changes is larger for potentially overinvesting firms than for value-maximizing firms. They use Tobin's Q to identify either overinvesting firms or value-maximizing firms. Specifically, they define a firm as an overinvestor when Tobin's Q <1, and value maximizer when Tobin's Q > 1. Lang and Litzenberger (1989) find that the average return for Q < 1 firms are significantly higher than Q > 1 firms. Their empirical result is in favor of the free cash flow hypothesis that dividend increases by overinvesting firms signal the firm's intention to mitigate the agency problem.

Yoon and Stark (1995) argue that dividend changes provide information about changes in the managers' misuse of excess cash under the free cash flow hypothesis. Therefore, the main task is to test the relationship between dividend changes and a firm's investment opportunities. They identify 3,748 dividend increases and 431 dividend decreases between 1969 and 1988. Besides Tobin's Q, they use the direction of insider trading as an additional proxy for investment opportunity. They find that there is no significant difference in the magnitude of stock price reactions in response to the dividend announcements between low Q and high Q firms, after controlling for the size of dividend change, the expected dividend yield, and the market value of the firm. Their finding is in contrast to that of Lang and Litzenberger (1989) who present that the absolute announcement returns are larger for Tobin's Q < 1 firms than for Tobin's Q > 1 firms. Nevertheless, they admit that their finding alone may not be sufficient enough to contradict the free cash flow hypothesis, because the control variables (e.g., the size of dividend change, the expected dividend yield, and the market value of the firm) used

³ Jensen, Michael. C. (1986), Agency costs of free cash flow, corporate finance, and takeovers. The American economic review, pp. 323-329

in their study are correlated with the firm's investment opportunities.

Lie (2000) asserts that contradictory evidence provided by Lang and Litzenberger (1989) versus Yoon and Stark (1995) could possibly result from the confounding effects of dividend change expectations and investment opportunities. Therefore, studies of special dividends and share repurchases may offer more appropriate tests of the free cash flow hypothesis. He investigates 7,417 regular dividend increase, 570 special dividends, and 207 repurchases from 1978 to 1993. The study mainly offers two following implications. Firstly, the payout announcements for repurchases and large special dividends are positively related to the stock market reactions for low Q firms, but this is not the case for high Tobin's firms. This result demonstrates that incremental cash payouts are beneficial for mitigating potential overinvestment problems. Secondly, there is no evidence suggesting that small special dividends and regular dividend increase help mitigate the overinvestment problem. The finding suggests that large special dividends and share repurchases could be perceived as a remedy to the potential free cash flow problem. However, there is little evidence supporting that regular cash dividend increases signal the firm's intention to mitigate the agency problem.

2.3 Information Asymmetry and Dividend Signaling Theory

Dividend signaling theory traces its root in the Lintner's (1956) dividend model. Lintner (1956) documents that firms generally set long-term target dividend payout ratio, and slowly adjust it in respond to earnings changes. Firms tend to increase dividend payments only when managers believe that future cash flows would be sufficient to support the higher dividend payments, and firms tend to avoid dividend cuts if at all possible.

Miller and Modigliani (1961) document "the information content of dividends" in a theoretical framework. They suggest that outside investors interpret a change in the dividend rate as a change in management's views of future profit prospects for the firm where a firm has adopted a long-established target payout ratio, and changes of investors' expectation about future earnings will reflect on changes of stock price⁴.

2.3.1 Dividend Signaling Models

Ross (1977) develops a one-period incentive-signaling model by grouping firms as either Type A or Type B firms. Specially, returns of Type A firms equal a, and returns of Type B firms equal b. These two types of firms cannot be differentiated at time zero

⁴ Miller, M.H. and Modigliani, F. (1961), *Dividend Policy, Growth, and The valuation of Shares*. The Journal of Business, Vol. 34, No.4, pp.411-433

even though a > b. Type A firms are capable of supporting an increased dividend payout without increasing the likelihood of bankruptcy, whereas Type B firms are not. Ross (1977) assumes that managers' compensation is positively related to their firms' market value, and they will be penalized if their firms go bankrupt. He also assumes that managers have more inside information of their firms' future cash flows. Managers of Type B firms are aware of their situations that any increase in dividend payout would send a false signal, and possibly cause their companies to go bankrupt eventually. So they have no incentives to increase dividends. As a consequence, investors view a dividend increase as coming from a Type A firm which is worthy of high valuation since any dividend increase in Type A firms will not increase the likelihood of its bankruptcy.

Bhattacharya (1979) develops a model that attempts to explain why many firms choose to pay dividends even if there is a tax disadvantage. He assumes that outside investors have imperfect information about a firm's profitability, and information, such as accounting data, is not fully reliable in assessing a firm's profitability. Although the tax rate on dividends is higher than on capital gains, Bhattacharya (1978) shows that investors are willing to pay a higher tax rate if they receive the favorable signal that the firm will have greater value in the future. Therefore, dividend is a useful signaling device for outside investor to evaluate a firm's future profitability.

Miller and Rock (1985) develop a model assuming that managers know more information about their firms' current earnings than outside investors. They propose that managers could try to eliminate the existing information asymmetry, which would reduce the need to send the costly signals to outside investors. But this method also involves dead-weight costs. Miller and Rock propose that dividends and financing are opposite sides of the same thing, i.e., financing is actually the negative dividends. They conclude that an unexpected earnings change has the same influence on firm returns as an unexpected dividend change. In addition, they state that current dividend payment trend rather than dividend itself is the basis on which the market predicts a firm's future earnings.

2.3.2 Empirical Evidence on Dividend Signaling

Empirical studies on dividend signaling primarily focus on two lines: one line of the research focuses on the dividend announcement effect by examining the abnormal returns around dividend announcement dates. The other line of the research focuses on examining the relationship between dividend changes and future earnings.

The empirical analysis of the dividend announcement effect is first highlighted by Pettit (1972). Pettit finds a positive pattern between dividend changes and abnormal stock returns surrounding the announcement date of the dividend change. Using a monthly and daily basis, Pettit finds that on average, the market makes unbiased value estimates from the information conveyed in these dividend change announcements and that stock price fully reflects the dividend change immediately on the announcement date or the following day.

However, Watts (1973) presents some evidence showing that the information content of dividends is trivial. Although he finds an association between unexpected dividend changes and future earnings changes, the average absolute size of future information conveyed by the dividend change is very small. He documents that traders cannot earn sufficient abnormal returns on such information content when transaction costs are considered.

Handjinicolaou and Kalay (1984) find that dividend increases do not have an impact on returns but that dividend decreases have a negative impact on returns. They investigate the bond price movements surrounding dividend announcements. They find that shareholder obtain gains when the information is positive. Bondholders, however, share losses when the information is negative. There exists a transfer of wealth from bondholders to shareholders when the dividend payment is altered. They document that their finding is consistent with the information content of dividends. On the other hand, Dhillon and Johnson (1994) cast doubt on the dividend signaling theory based on the fact that negative bondholder returns are associated with dividend increases. They conclude that dividend signaling might play a less important role in conveying information than it used to be.

Aharony and Swary (1980) examine quarterly cash dividends and earnings announcements made on different dates in a quarter. They find that changes in quarterly dividends provide information in addition to the quarterly earnings announcements. They find that stock prices react quickly to the dividend announcement, which supports the semi-strong form of efficient market hypothesis. Kane, Lee, and Marcus (1984) develop an expectation model of dividends and earnings based on a formula that determines what constitutes an unexpected change in dividend or earnings. They conclude that dividend announcement and earnings announcement have a significant impact on stock price individually.

In another line of the research that focuses on examining the relationship between dividend changes and future earnings, Ofer and Siegel (1997) document that analysts

revise their earnings expectations based on the unanticipated changes in dividend policy. They find a significant association between the size of earnings revision and the size of the unexpected dividend increase. Dyl and Weigand (1988) provide supplemental evidence that unexpected dividend changes can also affect the riskiness of earnings in addition to earnings expectations. They find that the riskiness of earnings, stock price variations, cash flows, and systematic risk of a firm drop significantly as the firm initiates dividends.

Healy and Palepu (1988) examine the relationship between dividend initiations and the level of earnings. They find that firms that initiate dividends typically have positive earnings changes in the year before and two years after the dividend initiation. This finding is consistent with Lintner's (1956) dividend model, indicating that dividend initiations are informative about future earnings growth. Conversely, they find that dividend omissions are associated with two years of earnings declines, and earnings continue to decline in the first subsequent year.

DeAngelo and DeAngelo (1990) examine the dividend policy adjustments of 80 NYSE firms that were experiencing financial distress between 1980 and 1985. They find a high incidence of dividend cuts by firms with persistent (three or more) losses, but no such pattern for firms with transitory losses. In other findings, DeAngelo, DeAngelo and Skinner (1996) study the signaling content of dividends for firms whose annual earnings experience a decline after having increased for at least nine consecutive years. They find that there is virtually no supporting evidence for the argument that dividend decisions help identify firms with superior future earnings. They document three factors that can help explain the phenomenon: managerial overoptimism, modest resource commitments, and managerial mistakes.

Benartzi, Michaely, and Thaler (1997) present that it is unable to document a reliable association between dividend changes and future earnings change. They document that firms increasing dividends have experienced significant earnings increases in the prior year, but show no subsequent unexpected earnings growth. Firms reducing dividends have experienced reductions in earnings in the prior year, but these firms show significant subsequent earnings increases. But they also document that dividend-increasing firms are less likely than non-changing firms to experience a drop in future earnings, which is consistent with Lintner's (1956) dividend model that firms increase dividends only if managers believe that earnings are increased permanently.

Nissim and Ziv (2001) try to figure out the relationship between dividend changes and future profitability. Assuming linear mean reversion in earnings changes, they find

a positive relationship between dividend changes and future earnings changes. One of their main findings is that the positive effect of dividend changes on future profitability are asymmetric between dividend increases and dividend decreases. Specifically, dividend increases are related with future profitability for at least four subsequent years, while dividend decreases are not significantly related to future profitability after controlling for current and expected profitability. However, Grullon, Michaely, Benartzi, and Thaler (2005) argue that the finding of Nissim and Ziv (2001) does not stand after controlling for the nonlinear patterns in the behavior of earnings.

In summary, the empirical evidence on the information content of dividend is mixed, with some studies support the signaling theory while the others contradict it. Dividend policy continues to be one of the controversial issues in corporate finance.

2.4 Firm Life Cycle Theory of Dividends

The life cycle theory of dividends is based on the notion that dividend policy is changing as the firm moves to a different stage of its life cycle. According to the life cycle of dividends, young firms typically do not choose to pay dividends because they face a relatively large investment opportunity set but are not sufficiently profitable to self-finance through internally generated capital as their capital spending is far beyond their profits. In addition, these firms face substantial hurdles in raising external funds from capital market. As a result, they will retain earnings as much as possible for future reinvestment. After a period of growth, these firms gradually reach the maturity stage when most of them have accumulated abundant profits. At this point, investment opportunity set is limited, growth rate and profitability have flattened, and systematic risk has declined. Finally, these mature firms begin distributing cash to shareholders. The firm life cycle theory of dividend demonstrates that trade-off between retention and distribution is different among different stages of a firm's life cycle.

So far, few studies have directly examined the life cycle theory of dividends but empirical evidence generally supports it. Fama and French (2001) investigate the time-series shift in the dividend payment behavior of publicly traded U.S. firms between 1926 and 1999. They find a substantial decline in the proportion of firms that pay dividends. They show that 66.5% of industrial firms paid dividends in the year 1978, while only 20.8% did so in 1999. They document that the declining incidence is due in part to the changing firm characteristics of new listed firms after 1978. These new lists are dominated by firms that have small size, low (or negative) profits and high growth rates, which are characteristics of never paying dividends. Their results indicate

that firm life cycle factors play an important role in the dividend decision. In particular, dividend-paying firms are typically large, more profitable. Their retained earnings are sufficient to cover capital spending. On the other hand, non-payers are small, less profitable and fast-growing. These firms have not generated ample profits to fully meet their financing needs and require external capital. In summary, this study shows that the decision to pay cash dividends is mainly dominated by firm characteristics that determine the firm's life cycle stage.

DeAngelo, DeAngelo and Stulz (2006) try to explicitly test the life cycle theory of dividends based upon Fama and French's (2001) study. They link the probability of a firm that pays dividends to the earned/contributed capital mix that is measured by the ratio of retained earnings to total common equity (RE/TE) or total assets (RE/TA). They assert that RE/TE (RE/TA) is a good proxy for a firm's life cycle stage. Low RE/TE (RE/TA) firms are in capital infusion stage, whereas high RE/TE (RE/TA) firms are in mature stage with substantial cumulative profits. Hence, they predict that the probability of a firm that pays dividends will increase with the RE/TE (RE/TA). Using a sample of publicly traded U.S. firms between 1972 and 2002, they find supporting evidence for the life cycle theory of dividends. They report a significantly positive relation between the proportion of publicly traded industrial firms that pay dividends and RE/TE (RE/TA), after controlling for firm current and lagged profitability, growth opportunities, size, total equity, cash balances and dividend history. Their finding also holds for dividend initiation and omission.

Skinner (2008) studies the firm payout policy including dividends and share repurchases. He finds that for firms that distribute cash through dividends and share repurchases, the magnitude of share repurchases is dominated by earnings over a two-or three-year window, which supports the life cycle theory. In addition, he finds that firms are increasingly using share repurchases instead of dividends to payout earnings.

Denis and Osobov (2008) extend the research to six developed countries. They study the cross-sectional determinants of dividend policy in U.S., Canada, U.K., Germany, France, and Japan over the period between 1989 and 2002. They document that the propensity to pay dividends is high among firms that are large, highly profitable, and retained earnings are a large portion of total equity. On the other hand, von Eije and Megginson (2008) investigate dividends and share repurchases of firms listed in 15 European Union nations from 1989 to 2005. However, they find no association between the likelihood of cash payouts and the ratio of retained earnings to total equity. But they do find that firm financial reporting frequency, size, age, and past profitability

are positively associated with cash payouts.

Recently, Brockman and Unlu (2011) document that retained earnings decile is positively related to the propensity to pay dividends in a global setting. They find that firms that initiate dividend payments increase their retained earnings decile rank prior to their dividend initiations, and firms that omit dividend payments decrease their retained earnings rank prior to their dividend omissions. These results are in line with the life cycle theory of dividends.

3. Data & Methodology

This section first provides the hypotheses development which I will examine in section 3.4. Then I discuss the variable construction, data selection criteria and model specifications in my study. The last part contains the empirical tests of the models where I will discuss the empirical results and whether my hypotheses are supported by the empirical tests.

3.1 Hypothesis Development

Based on the literature review above, I make the three following hypotheses for my study.

3.1.1 Hypothesis 1: Dividend Changes are Informative about Future Earnings.

According to the dividend signaling theory, if dividend policy changes, it is likely to affect future earnings. In other words, dividend changes are significantly related to earnings changes in each of the two years subsequent to the dividend change.

3.1.2 Hypothesis 2: Dividend Changes are Informative about Future Earning, Controlling for Past Earnings Changes and the Lagged ROE.

Benartzi et al. (1997) report that dividend changes are highly correlated with past earnings changes. Hence, the relation between dividend changes and future earnings changes may capture the autocorrelation in the earnings change series. Ohlson and Penman (1982) document that earnings to the book value of equity (ROE) is an important predictor of future earnings changes. High (low) ROE is expected to decrease (increase) future earnings due to the mean-reverting behavior of earnings changes. Dividend changes may be correlated with ROA as well. As a result, I include past earnings changes and the lagged ROE in the regression models as additional control variables. I therefore hypothesize that dividend changes are informative about future earnings, controlling for past earnings changes and the lagged ROE.

3.1.3 Hypothesis 3: Dividend Changes are Positively (Negatively) Related to Future Earnings Changes for the Corresponding (Contradictory) Sign Group.

Motivated by Koch and Sun (2004) who examine the market reactions to dividend change announcements by distinguishing dividend changes between confirmatory dividend changes (observations that the dividend change and recent past earnings change are of the same sign) and contradictory dividend changes (observations that the dividend change and recent past earnings change have opposite signs), I will additionally divide my firm-year observations that make non-zero dividend changes into two subgroups in the same way as Koch and Sun (2004). I hypothesize that confirmatory dividend changes are likely to be positively related to earnings changes in the two subsequent years. One of the possible explanations is that dividends are paid out with current or prior earnings. In the most case, change in dividends follows change in past earnings in the same direction. So these confirmatory dividend changes are likely to signal that the direction of past earnings change is expected to be sustainable in the future. Conversely, I hypothesize that contradictory dividend changes are likely to be negatively related to earnings changes in the two subsequent years. One of the possible explanations for the negative relation is that for firms with contradictory dividend decreases, managers may have seen the potential investment opportunities and intend to keep more funds for future investments which are likely to increase subsequent earnings, and that for firms with contradictory dividend increases, managers may want to satisfy equity investors through increasing dividends but less funds will be available for future development since they are paying more dividends in the context of earnings decreases, then their future earnings may get worse. Although there may be other reasons for these firms doing so, and different reasons may lead to different outcomes. In my study, I only hypothesize that contradictory dividend changes are likely to be negatively related to earnings changes in the two subsequent years and test whether my hypothesis can be supported by the empirical results.

3.2 Variable Construction

The key variables used in the study are earnings changes, dividend changes and return on equity. Firstly, earnings used in this study are earnings before extraordinary items because it is the usual case in many previous studies (Nissim and Ziv (2001) and Grullon et al., 2005)). The annual past earnings change is defined as follows:

 Δ PastEarnings_{it} = $(Earnings_{it} - Earnings_{i(t-1)})/BookEquity_{i(t-1)}$

I define past earnings change for firm i at time t as the difference in the annual earnings before extraordinary items between year t and t-1, deflated by the book value of equity at the beginning of the dividend change year. Nissim and Ziv (2001) suggest that it is better to deflate the earnings change by the book value of equity rather than the market value of equity. Since price reflects expectations about future earnings, the ratio of earnings to price is likely to be negatively related to the expected change in earnings (Penman (1996)⁵

Future earnings change for firm i at time t is defined as follows:

$$\Delta \text{FutureEarnings}_{i(t+1)} = (Earnings_{i(t+1)} - Earnings_{it}) / BookEquity_{i(t-1)}$$

$$\Delta \text{FutureEarnings}_{i(t+2)} = (Earnings_{i(t+2)} - Earnings_{i(t+1)}) / BookEquity_{i(t-1)}$$

The two equations represent the earnings change in the first and the second year following the dividend change respectively. The future earnings changes are also deflated by the book value of equity at the beginning of the dividend change year.

Dividend change for firm *i* at time *t* is measured by the rate of dividend change. The dividend change is defined as follows:

$$\Delta Div_{it} = (Div_{it} - Div_{i(t-1)})/Div_{i(t-1)}$$

Where Div_{it} is the dividend per share for firm i at time t, Div_{t-1} is the dividend per share for firm i at time t-1. Clearly, two consecutive years of dividend payments are necessary for the calculation of the dividend change.

The lagged return on equity is defined as earnings before extraordinary items scaled by the book value of equity. I use the return on equity in year t and t+1 in my analysis, they are defined as follows:

$$ROE_{it} = Earnings_{it}/BookEquity_{it}$$

 $ROE_{i(t+1)} = Earnings_{i(t+1)}/BookEquity_{i(t+1)}$

3.3 Dataset

The sample consists of firms recorded in Compustat annual file over the sample period 1993-2011. To be included in the sample, a firm must meet the following criteria: (1) A

⁵ Nissim, D. and Ziv, A. (2001), *Dividend Changes and Future Profitability*. The Journal of Finance, vol. 56, No.6, pp.2111-2133

firm must distribute cash dividend (US dollars) for at least two consecutive years because two consecutive years of dividend payments are necessary for the calculation of the change in dividends. This means that dividend initiation and omission events are excluded from the study. (2) The sample is restricted to industrial firms listed in NYSE, AMEX and NASDAQ. Consistent with most of the previous literatures (e.g. Benartzi et al. (1997) and Nissim and Ziv (2001)), financial industry (SIC codes between 6000 and 6999) and utilities (SIC codes between 4900 and 4949) are excluded from the sample. (3) There are no missing records of stock returns and market returns to estimate the cumulative abnormal return. The event window for my event study is [-2, 2], and the estimation window is [-230,-30]. Stock returns are available in the CRSP file. (4) I do not investigate stock dividends, stock repurchase or special dividends since these dividend behaviors do not possess the same level of information content as regular cash dividends changes.

My sample consists of 7,192 dividend increases, 2,247 dividend decreases and 2,752 no-change observations. They are made by 1,146 industrial firms listed in NYSE, AMEX and NASDAQ between 1993 and 2011. Table 1 presents the descriptive statistics of the sample. Panel A shows the total number and percentage of firm-year observations in different dividend groups. Dividend increases account for the largest proportion in my sample, with 59 percent of the firm-year observations increasing dividends. It is in line with previous studies (e.g. DeAngelo and DeAngelo (1990) and Nissim and Ziv (2001)) that dividend increases are the most frequent cases. 18 percent of the firm-year observations experience dividend cuts, which are far less frequent than dividend increases.

Panel B reports values of mean, standard deviation, 10th, 25th, 50th, 75th, and 90th percentiles of dividend changes (ΔDiv_{it}) for dividend increases, dividend decreases, and all dividend changes. The unconditional mean (median) of dividend changes is 52.1 percent (12.5 percent) for dividend increases, -36.8 percent (-35.3 percent) for dividend decreases and 30.9 percent (8.3 percent) for all dividend changes. The mean dividend increases is larger in magnitude than the mean dividend decreases, whereas the median dividend increases is smaller in magnitude than the median dividend decreases. It means that the mean dividend increases may be affected by extremely large dividend increases.

Panel C shows the mean and median values of cumulative abnormal returns around dividend change announcements. The event window for calculating cumulative abnormal returns is [-2, 2], and the estimation windows for estimating normal returns

is [-230, -30]. I use value-weighted returns for NYSE, AMEX and NASDAQ as proxy for market return, and estimate the following market model over the estimation window for each dividend change announcement:

$$R_{it} = \alpha + \beta R_{mt} + \varepsilon_{it}$$

Where R_{it} is the stock return of firm i at time t, and R_{mt} is the value-weighted return for NYSE, AMEX and NASDAQ at time t. The abnormal returns and cumulative abnormal returns in the event window are calculated as follows:

$$AR_{it} = R_{it} - NR_{it} = R_{it} - (\hat{\alpha} + \hat{\beta}R_{mt})$$

$$CAR_i = \sum_{t=t_1}^{t=t_2} AR_{it}$$

Where AR_{it} is the abnormal return of firm i at time t, NR_{it} is the normal return that is calculated by the estimated market model, CAR_i is the cumulative abnormal returns of firm i. $[t_1, t_2]$ is the event window.

In order to test whether the cumulative abnormal returns are significantly different from zero. I use t-tests to test the mean cumulative abnormal returns, and Wilcoxon signed-rank tests to test the median cumulative abnormal returns. The mean (median) cumulative abnormal returns (the cumulative average abnormal return) during the two days surrounding the announcement date of dividend change is 0.95 percent (0.57 percent) for dividend increases, and -0.32 percent (-0.13 percent) for dividend decreases. They are all significantly different from zero except the median cumulative abnormal return for dividend decreases. Consistent with various previous studies (e.g., Kalay and Loewenstein (1985)), the results in Panel C show a positive relation between dividend changes and abnormal stock returns. On average, Investors can exploit significantly positive stock returns when a firm announces a dividend increase.

Table 1 Sample Descriptive Statistics

The sample consisits of 12,191 firm-year observations by 1,146 industrial firms that are listed in NYSE, AMEX and NASDAQ between 1993 and 2011. Financial industry (SIC codes between 6000 and 6999) and utilities industry (SIC codes between 4900 and 4949) are excluded from the sample. Firms in sample must have paid cash dividends for at least two consecutive years. Panel A shows the total number and percentage of firm-year observations in different dividend groups. Panel B shows descriptive statistics for dividend changes. The variable ΔDiv_{it} is the change in cash dividend per share in the dividend change year, divided by the previous year's level. Panel B shows values of mean, standard deviation, 10th, 25th, 50th, 75th, and 90th percentiles for the variable ΔDiv_{it} in the different dividend change groups. Panel C shows the mean and median cumulative abnormal returns around dividend change announcements for dividend increases and decreases. The event window to calculate the abnormal returns is [-2, 2]. For the mean column, I use t-test to test whether the means are significantly different from zero. For the median column, I use Wilcoxon signed-rank test to test whether the medians are significantly different from zero. The critical values of the mean and median tests are reported in parentheses.

Panel A: Sample							
	Divide	d Increase	s No C	hange	Dividend	Decreases	Total
Firm-year	7,192		2,752	2	2,24	1	2,191
Observations							
Percentage	59%		23%		18%	1	00%
Panel B: Sample Des	criptive St	atistics for	r Dividen	d Changes	S		
	Mean	SD	P10	P25	P50	P75	P90
	Dividend	l Increases	s (N=7,19	(2)			
ΔDiv_{it}	0.521	3.082	0.029	0.063	0.125	0.272	0.714
	Dividend	l Decrease	es (N=2,2	47)			
ΔDiv_{it}	-0.368	0.230	-0.722	-0.500	-0.353	-0.188	-0.081
	All Divid	dend Chan	iges (N=9	,439)			
ΔDiv_{it}	0.309	2.719	-0.413	0.000	0.083	0.200	0.529
Panel C: Cumulative	Abnormal	Return ar	ound Ann	ounceme	nts of Divid	end Chang	e
				CAR_i			
				Mean		Median	
Dividend Increases				0.95%*	**	0.57%**	*
				(16.95)		(15.93)	
Dividend Decreases				-0.32%	**	-0.13%	
				(-2.05)		(-1.42)	

^{***} Statistical significance at the 1 percent level ** Statistical significance at the 5 percent level

3.4 Methodology & Empirical Results

3.4.1 Model Specifications

Based on the theoretical and empirical studies on the information content of dividend hypothesis, it is important to test whether a firm that increases (decreases) dividend payments in a year will have an unexpected increase (decrease) in future earnings.

Following Nissim and Ziv (2001), I use model (1) and (2) to test hypothesis 1.

$$\Delta \text{FutureEarnings}_{i(t+1)} = \alpha_0 + \alpha_1 \Delta Div_{it} + \varepsilon_{i(t+1)}$$
 (1)

$$\Delta \text{FutureEarnings}_{i(t+2)} = \alpha_0 + \alpha_1 \Delta Div_{it} + \varepsilon_{i(t+2)}$$
 (2)

Where Δ FutureEarnings $_{i(t+1)}$ is the change in earnings in the first year after the dividend change, Δ FutureEarnings $_{i(t+2)}$ is the change in earnings in the second t year after the dividend change. ΔDiv_{it} is the rate of the dividend change in the dividend change year.

To control for the autocorrelation in earnings change series and correlation between dividend changes and the lagged return on equity, I include past earnings changes and the ratio of earnings to the book value of equity in model (3) and (4). In addition, I distinguish between dividend increase and dividend decrease since some previous studies (e.g. DeAngelo and DeAngelo (1990) and Benartzi et al. (1997)) document that the effect of dividend changes on future earnings changes is asymmetric for dividend increases and decreases. Therefore, dividend increases and decreases are distinguished by introducing the dummy variables.

I estimate the following model (3) and (4) to test hypothesis 2:

$$\Delta \text{FutureEarnings}_{i(t+1)} = \alpha_0 + \alpha_1 PosDum_{it} \cdot \Delta Div_{it} + \alpha_2 NegDum_{it} \cdot \Delta Div_{it}$$

$$+ \alpha_3 \Delta \text{PastEarnings}_{it} + \alpha_4 \text{ROE}_{it} + \varepsilon_{i(t+1)} \tag{3}$$

$$\Delta \text{FutureEarnings}_{i(t+2)} = \alpha_0 + \alpha_1 PosDum_{it} \cdot \Delta Div_{it} + \alpha_2 NegDum_{it} \cdot \Delta Div_{it}$$

$$+ \alpha_3 \Delta \text{PastEarnings}_{it} + \alpha_4 \text{ROE}_{i(t+1)} + \varepsilon_{i(t+2)} \tag{4}$$

Where $PosDum_{it}$ is a dummy variable that equals 1 if dividend changes are positive in year t and 0 otherwise. $NegDum_{it}$ is also a dummy variable that equals 1 if dividend changes are negative in year t and 0 otherwise. $\Delta PastEarnings_{it}$ is the change in earnings in the dividend change year. ROE_{it} is the return on equity in the dividend change, and $ROE_{i(t+1)}$ is the return on equity in the first year after the

dividend change. All the other variables are defined in the same way as in model (1) and (2).

3.4.2 Empirical Tests of the Models

Table 2 reports the empirical results of model (1) and (2) to test hypothesis 1 that dividend changes are informative about future earnings. The results show that the coefficients on dividend changes (α_1) in model (1) and (2) are both statistically insignificant, indicating that dividend changes do not contain information about earnings changes in each of the two years subsequent to the dividend change. Noticeably, the R-squares are zero in Table 2. This means that the variable of dividend changes alone provides virtually no information about future earnings changes based on the whole sample data. These results are consistent with the previous study carried out by Benartzi et al. (1997) who also find no significant relation between dividend changes and future earnings. These results are consistent with the previous study carried out by Benartzi et al. (1997) who also find no significant relation between dividend changes and future earnings changes, but inconsistent with the results of Nissim and Ziv (2001) who find that dividend changes are positively related to earnings changes in each of the two years following the dividend change. Nissim and Ziv (2001) attribute the insignificant results of Benartzi et al. (1997) to the measurement error in the dependent variable because Benartzi et al. (1997) deflate earnings changes by the market value of equity. Nissim and Ziv (2001) document that stock price also incorporates expectations about future earnings, it is likely that the earnings-to-price ratio is negatively associated with the price-deflated earnings change in the first year subsequent to the dividend change (Penman (1996)). To avoid any possible distortions from the deflator, they suggest it is better to deflate the earnings changes by the book value of equity rather than by the market value of equity. I deflate the earnings changes by the book value of equity as suggested by Nissim and Ziv (2001), but I still find no significant relation between dividend changes and future earnings changes. Therefore, hypothesis 1 is not confirmed.

Table 3 shows the empirical results of model (3) and (4) to test hypothesis 2 that dividend changes are informative about future earnings changes, after controlling for past earnings changes and the lagged return on equity which are expected to be correlated with dividend changes. 6 The results in Table 3 indicate that the coefficients on dividend increases and decreases are all insignificant, showing that both dividend increases and dividend decreases are still unrelated to earnings changes

-

⁶ Surprisingly, the correlation between dividend changes and past earnings changes is 0.001, and the correlation between dividend changes and the current return on equity is also approximately 0.001 in my sample.

in each of the two years subsequent to the dividend change based on the whole sample data, even controlling for past earnings changes and the lagged ROE. The coefficient on past earnings changes (α_3) in model (3) is negative and statistically significant at the 5 percent level. This indicates that past earning changes are negatively related to earnings changes in the first year subsequent to the dividend change based on the whole sample data. However, no significant relation is found between past earnings changes and earnings changes in the second year after the dividend change since the coefficient on the earnings changes (α_3) in model (4) is insignificant. For the lagged return on equity, the coefficient on the lagged return on equity (α_4) is positive and significant at 5 percent level in Model (4) but insignificant in Model (3). This indicates that the lagged return on equity is positively related to earnings changes in the second year rather than the first year following the dividend change. Hence, Ohlson and Penman's (1982) finding which suggests that high (low) ROE implies an expected decrease (increase) in future earnings is unconfirmed.

Therefore, hypothesis 2 is unconfirmed as well because neither dividend increases nor dividend decreases are found to be significantly related to earnings changes in each of the two subsequent years.

To test hypothesis 3, I use the 9,438 firm-year observations by 1,142 industrial firms that have made non-zero dividend changes from 1993 to 2011. Following Koch and Sun (2004), I divide my sample into two subgroups based on whether the dividend change and past earnings change are of the same sign. The number of firms with confirmatory dividend changes is 5,530, and the number of firms with contradictory dividend changes is 3,908.

The regression results of model (1) and (2) in which the variable of dividend changes is the only explanatory variable are not reported here. The coefficients on dividend changes (α_1) remain insignificant for both confirmatory and contradictory groups. It indicates that when future earnings changes are explained by dividend changes alone, dividend changes are unrelated to earnings changes in each of the two subsequent years.

Table 4 shows the regression results of model (3) and (4) in which dividend increases and dividend decreases are distinguished by including dummy variables, and past earnings changes and the lagged return on equity are added as additional control variables. Panel A shows the results for firms with confirmatory dividend changes. The coefficient on dividend increases (α_1) in model (3) is positive and significant at the 10 percent level, showing that confirmatory dividend increases are positively related to

earnings changes in the first subsequent year. However, no significant relation is found between confirmatory dividend increases and earnings changes in the second subsequent year because the coefficient on dividend increases (α_1) in model (4) is insignificant. On the other hand, the coefficients on dividend decreases (α_2) in model (3) and (4) for firms with confirmatory dividend changes are both positive and highly significant at the 1 percent level. It means that when dividend decreases are preceded by past earnings decreases, dividend decreases are positively related to earnings changes in each of the two years after the dividend change. Moreover, the coefficients on dividend decreases are substantially larger in magnitude compared to those on dividend increases, indicating that dividend decreases provides more information content about future earnings changes. As for the economic significance of dividend decreases, one standard deviation (0.254) decrease in negative dividend changes leads to a decrease in earnings changes in the first year following the dividend change by 9.3 percent, and a decrease in earnings changes in the second year following the dividend change by 29.0 percent.

In Panel B of Table 4 where firms make contradictory dividend changes, the coefficients on dividend increases (α_1) are both insignificant, showing that dividend increases provide little information about future earnings changes. The coefficient on dividend decreases (α_2) in model (4) is negative and significant at the 10 percent level. It means that when dividend decreases are preceded by past earnings increases, dividend decreases are negatively related to earnings changes in the second year after the dividend changes. This evidence is consistent with the second part of hypothesis 3 where I hypothesize that dividend changes are negatively related to future earnings changes for firms with contradictory dividend changes. There may be several reasons that can explain the negative relation between contradictory dividend decreases (i.e. dividend cuts are preceded by past earnings increases) and earnings changes in the second year after the dividend decrease. The first possible explanation is that some managers may have seen the potential investment opportunities and intend to retain more funds for future investments which are likely to increase earnings in the subsequent years. The second possible explanation is that dividend policy is a tool for changing financing mix as Damodaran A. (2010, pp 10.44) states. ⁷ Increasing dividends increases a firm's financial leverage, and decreasing dividends reduces a firm's financial leverage. So a firm that decreases dividends may want to satisfy current bondholders and attract potential bondholders because less wealth is transferred from lenders to shareholders. Also, firms with lower financial leverage are

_

⁷ Damodaran A. (2010), *Applied Corporate Finance*. John Wiley & Sons, Chapter 10, pp.44

capable of raising debt capital at a lower cost. Therefore, if it is the case, firms are likely to increase earnings in the subsequent years. The third possible explanation is that some managers may have realized the changing competition environment and want to hoard more cash to strengthen their competitive outcomes. Fresard (2010) documents that large cash holdings lead to significant future market share gains at the expense of industry rivals. In addition to these possible explanations, firms may have other reasons to decrease dividends after the past earnings increase.

In sum, the empirical results in Table 5 indicate that confirmatory dividend increases and decreases are both positively related to earnings changes in each of the two years after the dividend change, and that only contradictory dividend decreases are negatively related to earnings changes in the second subsequent year. However, contradictory dividend increases are unrelated to earnings changes in each of the two subsequent years, and contradictory dividend decreases are unrelated to earnings changes in the first subsequent year. Overall, the significant results in Table 4 are in line with hypothesis 3, while the insignificant results remain unconfirmed.

Table 2

Results from Regressions of Future Earnings Change on the Dividend Change

This table reports the regression results of model (1) and (2).

Model (1)
$$\Delta$$
FutureEarnings $_{i(t+1)} = \alpha_0 + \alpha_1 \Delta Div_{it} + \varepsilon_{i(t+1)}$

Model (2)
$$\Delta$$
FutureEarnings $_{i(t+2)} = \alpha_0 + \alpha_1 \Delta Div_{it} + \varepsilon_{i(t+2)}$

The sample consists of 12,191 firm-year observations by 1,146 industrial firms that are listed in NYSE, AMEX and NASDAQ between 1993 and 2011. Financial industry (SIC codes between 6000 and 6999) and utilities industry (SIC codes between 4900 and 4949) are excluded from the sample. Firms in sample must have paid cash dividends for at least two consecutive years. The variable Δ FutureEarnings $_{i(t+1)}$ and Δ FutureEarnings $_{i(t+2)}$ represent the change in the earnings before extraordinary items for the first year and the second year respectively subsequent to the dividend change, scaled by the book value of equity at the beginning of the dividend change year. The variable ΔDiv_{it} is the rate of dividend change, defined as the change in dividend per share scaled by the dividend per share in a prior year. The t-statistics are reported in parentheses.

	α_0	α_1	R^2	N
Model (1)	0.033	-0.001	0.000	12,155
Model (2)	(2.58) 0.015	(-1.12) -0.003	0.000	12,091
	(0.74)	(-1.46)		,

Table 3

Results from Regressions of Future Earnings Change on the Dividend Change and other Control Variables

This table reports the regression results of model (3) and (4).

$$\Delta \text{FutureEarnings}_{i(t+1)} = \alpha_0 + \alpha_1 PosDum_{it} \cdot \Delta Div_{it} + \alpha_2 NegDum_{it} \cdot \Delta Div_{it}$$

$$+ \alpha_3 \Delta \text{PastEarnings}_{it} + \alpha_4 \text{ROE}_{it} + \varepsilon_{i(t+1)}$$
(3)

 $\Delta \text{FutureEarnings}_{i(t+2)} = \alpha_0 + \alpha_1 PosDum_{it} \cdot \Delta Div_{it} + \alpha_2 NegDum_{it} \cdot \Delta Div_{it}$

$$+\alpha_3 \Delta \text{PastEarnings}_{it} + \alpha_4 \text{ROE}_{i(t+1)} + \varepsilon_{i(t+2)}$$
 (4)

The sample consists of 12,191 dividend events made by 1,146 industrial firms that are listed in NYSE, AMEX and NASDAQ between 1993 and 2011. Financial industry (SIC codes between 6000 and 6999) and utilities industry (SIC codes between 4900 and 4949) are excluded from the sample. Firms in sample must have paid cash dividends for at least two consecutive years. The variable Δ FutureEarnings $_{i(t+1)}$ and Δ FutureEarnings $_{i(t+2)}$ represent the change in the earnings before extraordinary items for the first year and the second year respectively subsequent to the dividend change, scaled by the book value of equity at the beginning of the dividend change year. The variable ΔDiv_{it} is the rate of dividend change, defined as the change in dividend per share scaled by the dividend per share in a prior year. The variable Δ PastEarnings_{it} is the change in earnings before extraordinary items in the dividend change year, scaled by the book value of equity at the beginning of the dividend change year. The variable $PosDum_{it}$ ($NegDum_{it}$) is a dummy variable that equals 1 if the dividend change is positive (negative) and 0 otherwise. The variable ROE_{it} ($ROE_{i(t+1)}$) represent the return on equity in the dividend change year (one year) after the dividend change, defined as earnings before extraordinary items in year t (t+1) scaled by the book value of equity in year t (t+1)1). The *t*-statistics are reported in parentheses.

	α_0	α_1	α_2	α_3	α_4	R^2	N
Model(3)	0.029 (2.26)	-0.001 (-0.98)		-0.037** (-1.97)	0.001 (1.54)	0.019	12,145
Model(4)	-0.001 (-0.10)	-0.001 (-1.15)	-0.229 (-1.60)	0.004 (0.13)	0.002** (2.26)	0.034	12,071

^{**}statistically significance at the 5 percent level

Table 4

Results from Regressions of Future Earnings Change on the Dividend Change and other Control Variables for Two Subgroups

This table reports the regression results of model (3) and (4).

Model (3)
$$\Delta$$
FutureEarnings $_{i(t+1)} = \alpha_0 + PosDum_{it} \cdot \Delta Div_{it} + \alpha_2 NegDum_{it} \cdot \Delta Div_{it} + \alpha_3 \Delta PastEarnings_{it} + \alpha_4 ROE_{it} + \varepsilon_{i(t+1)}$

Model (4)
$$\Delta$$
FutureEarnings $_{i(t+2)} = \alpha_0 + PosDum_{it} \cdot \Delta Div_{it} + \alpha_2 NegDum_{it} \cdot \Delta Div_{it} + \alpha_3 \Delta PastEarnings_{it} + \alpha_4 ROE_{i(t+1)} + \varepsilon_{i(t+2)}$

The variable Δ FutureEarnings $_{i(t+1)}$ and Δ FutureEarnings $_{i(t+2)}$ represent the change in the earnings before extraordinary items for the first year and the second year respectively subsequent to the dividend change, scaled by the book value of equity at the beginning of the dividend change year. The variable ΔDiv_{it} is the rate of dividend change, defined as the change in dividend per share scaled by the dividend per share in a prior year. The variable $\Delta PastEarnings_{it}$ is the change in earnings before extraordinary items in the dividend change year, scaled by the book value of equity at the beginning of the dividend change year. The variable $PosDum_{it}$ ($NegDum_{it}$) is a dummy variable that equals 1 if the dividend change is positive (negative) and 0 otherwise. The variable ROE_{it} ($ROE_{i(t+1)}$) represent the return on equity in the dividend change year (one year) after the dividend change, defined as earnings before extraordinary items in year t (t + 1) scaled by the book value of equity in year t (t + 1). The t-statistics are reported in parentheses.

Panel A: Results for the observations with confirmatory dividend changes							
	$lpha_0$	$lpha_1$	α_2	α_3	α_4	R^2	N
Model (3)	0.092	0.003*	0.368***	-1.405***	-0.001	0.772	5,518
	(5.52)	(1.66)	(2.81)	(-5.49)	(-0.05)		
Model (4)	0.222	0.007	1.141***	-3.612***	0.008	0.740	5,487
	(3.65)	(1.39)	(3.01)	(-3.88)	(1.48)		
Panel B: Res	ults for the o	bservations v	vith contradict	ory dividend	changes		
	α_0	α_1	α_2	α_3	α_4	R^2	N
Model (3)	0.065	-0.001	0.128	-0.027***	0.002	0.014	3,889
	(1.42)	(-0.87)	(1.34)	(-2.92)	(0.43)		
Model (4)	-0.032	0.000	-0.18	0.030	-0.021	0.020	3,864
	(-0.79)	(0.25)	(-1.65)	(3.72)	(-1.09)		

^{***} statistical significance at the 1 percent level * statistical significance at the 10 percent level

4. Robustness Checks

This section presents the robustness checks of my empirical results showed in Section 3.4.2. Instead of deflating changes in earnings by the book value of equity at the beginning of the dividend change year, I deflate them by the book value of total assets at the beginning of the dividend change year.

Table 5 presents the regression results for model (1) and (2) when changes in earnings are deflated by the book value of total assets at the beginning of the dividend change year. Similar to Table 2, none of the coefficients on dividend changes (α_1) are significant. Based on the whole sample data, dividend changes are unrelated to earnings changes in the two subsequent years when earnings changes are deflated by the book value of total assets. The R-squares are also close to zero, showing that dividend changes provides little information about future earnings changes when the variable of dividend changes is the only explanatory variable in the models.

Table 6 presents the regression results for model (3) and (4) when changes in earnings are deflated by the book value of total assets at the beginning of the dividend change year. Similar to Table 3, none of the coefficients on either dividend increases or dividend decreases (α_1 and α_2) are significant. Based on the whole sample, dividend changes are unrelated to earnings changes in the two subsequent years when earnings changes are deflated by the book value of total assets. Past earnings changes show a negative relation with earnings changes in each of the two years following the dividend change as the coefficients on past earnings changes (α_3) remain negative and statistically significant at the 1 percent level in both model (3) and model (4).

I also perform the robustness checks for firms with confirmatory dividend changes and firms with contradictory dividend change though the results are not reported here. When changes in earnings are deflated by the book value of total assets at the beginning of the dividend change year, confirmatory dividend decreases are positively related to earnings changes in the first subsequent year, and contradictory dividend decreases are negatively related to earnings changes in the first subsequent year. The other coefficients on either dividend increases or dividend decreases are insignificant.

The robustness check results show that dividend changes are consistently unrelated to earnings changes in each of the two subsequent year based on the whole sample data. Also, the positive relation between confirmatory dividend changes and earnings changes in the two subsequent years is no more significant in the robustness

checks. Only confirmatory dividend decreases are positively related to earnings changes in the second subsequent year. To conclude, when earnings changes are deflated by the book value of total assets rather than the book value of equity in the dividend change year, dividend changes offer little information about future earnings changes.10

Table 5

Results from Regressions of Future Earnings Change on the Dividend

Change by Alternative Deflator

This table reports the regression results of model (1) and (2).

Model (1)
$$\Delta$$
FutureEarnings $_{i(t+1)} = \alpha_0 + \alpha_1 \Delta Div_{it} + \varepsilon_{i(t+1)}$

The sample consists of 12,191 firm-year observations by 1,146 industrial firms that are listed in NYSE, AMEX and NASDAQ between 1993 and 2011. Financial industry (SIC codes between 6000 and 6999) and utilities industry (SIC codes between 4900 and 4949) are excluded from the sample. Firms in sample must have paid cash dividends for at least two consecutive years. The variable Δ FutureEarnings $_{i(t+1)}$ and Δ FutureEarnings $_{i(t+2)}$ represent the change in the earnings before extraordinary items for the first year and the second year respectively subsequent to the dividend change, scaled by the book value of equity at the beginning of the dividend change year. The variable ΔDiv_{it} is the rate of dividend change, defined as the change in dividend per share scaled by the dividend per share in a prior year. The t-statistics are reported in parentheses.

	α_0	α_1	R^2	N
Model (1)	0.07	-0.001	0.000	12,172
	(8.80)	(-1.19)		
Model (2)	0.006	-0.001	0.001	12,108
	(6.81)	(-1.64)		

33

¹⁰ In addition, the empirical results remain similar when I include year fixed effects, similar to Braggion & Moore (2011).

Table 6

Results from Regressions of Future Earnings Change on the Dividend Change and other Control Variables by Alternative Deflator

This table reports the regression results of Model (3) and (4).

Model (3)
$$\Delta$$
FutureEarnings $_{i(t+1)} = \alpha_0 + PosDum_{it} \cdot \Delta Div_{it} + \alpha_2 NegDum_{it} \cdot \Delta Div_{it} + \alpha_3 \Delta PastEarnings_{it} + \alpha_4 ROE_{it} + \varepsilon_{i(t+1)}$

Model (4)
$$\Delta$$
FutureEarning $s_{i(t+2)} = \alpha_0 + PosDum_{it} \cdot \Delta Div_{it} + \alpha_2 NegDum_{it} \cdot \Delta Div_{it} + \alpha_3 \Delta PastEarning $s_{it} + \alpha_4 ROE_{i(t+1)} + \varepsilon_{i(t+2)}$$

The sample consists of 12,191 firm-year observations by 1,146 industrial firms that are listed in NYSE, AMEX and NASDAQ between 1993 and 2011. Financial industry (SIC codes between 6000 and 6999) and utilities industry (SIC codes between 4900 and 4949) are excluded from the sample. Firms in sample must have paid cash dividends for at least two consecutive years. The variable Δ FutureEarnings $_{i(t+1)}$ and Δ FutureEarnings $_{i(t+2)}$ represent the change in the earnings before extraordinary items for the first year and the second year respectively subsequent to the dividend change, scaled by the book value of total assets at the beginning of the dividend change year. The variable ΔDiv_{it} is the rate of dividend change, defined as the change in dividend per share scaled by the dividend per share in a prior year. The variable Δ PastEarnings_{it} is the change in earnings before extraordinary items in the dividend change year, scaled by the book value of total assets at the beginning of the dividend change year. The variable $PosDum_{it}$ ($NegDum_{it}$) is a dummy variable that equals 1 if the dividend change is positive (negative) and 0 otherwise. The variable ROE_{it} ($ROE_{i(t+1)}$) represent the return on equity in the dividend change year (one year) after the dividend change, defined as earnings before extraordinary items in year t (t+1) scaled by the book value of equity in year t (t+1)1). The *t*-statistics are reported in parentheses.

	α_0	α_1	α_2	α_3	α_4	R^2	N
Model(3)	0.007	-0.001	-0.009	-0.241***	0.001	0.039	12,151
	(10.42)	(-0.20)	(-1.05)	(-6.54)	(1.10)		
Model(4)	0.006 (6.94)	-0.001 (-1.27)	-0.011 (-1.36)	-0.150*** (-3.76)	0.001 (0.66)	0.011	12,081

^{***} statistical significance at the 1 percent level

5. Summary and Recommendations

This section provides summary and recommendations for my study. Section 5.1 presents the results and conclusions for my three main hypotheses. Section 5.2 provides the implications and recommendations based on the results of my study.

5.1 Summary of the Research

This study investigates the information content of dividends by analyzing the impact of dividend changes on future earnings changes. Following Nissim and Ziv (2001), I examine the relationship between dividend changes and earnings changes in each of the two years following the dividend change. Also, motivated by Koch and Sun (2004) who examine the market reactions to dividend change announcements by distinguishing dividend changes between confirmatory dividend changes and contradictory dividend changes, I divide the firm-year observations that have made non-zero dividend changes into two subgroups based on whether the dividend change and past earnings change are of the same sign. I hypothesize that dividend changes are likely to be positively related to earnings changes in the two subsequent years for firms with confirmatory dividend changes, and negatively related to earnings changes in the two subsequent years for firms with contradictory dividend changes.

Many researches (e.g. Pettit (1972), Kaly and Loewenstein (1985)) document the information content of dividends by examining the relationship between dividend changes and abnormal stock returns around the dividend change announcement. I show that the mean (median) cumulative abnormal returns during the two days surrounding the announcement date of dividend change is 0.95 percent (0.57 percent) for dividend increases, and -0.32 percent (-0.13 percent) for dividend decreases, all statistically different from 0 (except the median cumulative abnormal returns for dividend decreases). These results are consistent with the previous studies which document that the abnormal stock returns are positive for dividend increases, and negative for dividend decreases.

Empirical results indicate that dividend changes provide virtually no information about earnings changes in the two subsequent years. All the coefficients on dividend changes in model (1) and (2) remain statistically insignificant, showing that dividend changes are unrelated to earnings changes in each of the two years following the dividend change in the simplest models in which the variable of dividend changes is the

only explanatory variable. Thus, hypothesis 1-Dividend changes are informative about future earnings changes- is unconfirmed.

After incorporating past earnings change and the lagged return on equity into the models and making a distinction between dividend increases and dividend decreases, I find that neither dividend increases nor dividend decreases are significantly related to earnings changes in the two subsequent years. Thus, hypothesis 2-Dividend changes are informative about future earnings changes, controlling for past earnings changes and the lagged return on equity- is unconfirmed as well.

When dividend changes are classified into confirmatory dividend changes and contradictory dividend changes, confirmatory dividend increases and decreases are both positively related to earnings changes in each of the two subsequent years. Contradictory dividend decreases are negatively related to earnings changes in the second subsequent year. However, these results are no more significant in the robustness checks. When changes in earnings are deflated by the book value of assets in the dividend change year, only the relation between contradictory dividend decreases and earnings changes in the first subsequent year is found to be significant.

When changes in earnings are deflated by the book value of equity, confirmatory dividend increases and decreases are positively related to earnings changes in each of the two subsequent years, and confirmatory dividend decreases are negatively related to earnings changes in the second subsequent year. When changes in earnings are deflated by the book value of total assets, contradictory dividend decreases are negatively related to earnings changes in the first subsequent year. Nevertheless, dividend decreases appear to offer more information about future earnings changes than dividend increases in terms of the size of the coefficients and the associated *t*-statistics, though not all the coefficients on dividend decreases remain significant.

One of the possible explanations for the positive relation between confirmatory dividend changes and earnings changes in the two subsequent years is that dividends are paid out with current or prior earnings. In the most case, change in dividends follows change in past earnings in the same direction. So these confirmatory dividend changes are likely to signal that the direction of change in past earnings is expected to be sustainable in the future. There may be several reasons that can explain the negative relation between contradictory dividend decreases (i.e. dividend cuts are preceded by past earnings increases) and earnings changes in the second year after the dividend decrease. Some managers may have seen better investment opportunities, or some managers may want to adjust their capital structure to attract more bondholders, or

some managers may want to strengthen their future market shares in response to the changing competition environment. Also, there may be other plausible reasons than can help explain the results. Thus, hypothesis 3- Dividend changes are positively related to future earnings changes for firms with confirmatory dividend changes, and negatively related to future earnings changes for firms with contradictory dividend changes- is also only partially confirmed as the results are not robust to an alternative measurement of the deflator.

5. 2 Recommendations

In summary, this study finds limited evidence supporting the information content of dividends. The empirical results indicate that variations in future earnings can be largely predicted by past earnings changes because past earnings changes are negatively related to earnings changes in each of the two subsequent years in most cases. In contrast to Nissim and Ziv (2001) who find that dividend changes are positively related to earnings changes in each of the two subsequent years, I find that dividend changes provide substantially limited information about a firm's future earnings

This study is based on industrial firms listed on the NYSE, AMEX, and NASDAQ from 1993 to 2011. It provides investors and equity market some understanding of whether dividend changes are informative about future earnings changes for U.S. publicly traded firms during the recent twenty years from 1993 to 2011. The results show that dividend changes provide little information about future earnings change.

Noticeably, the coefficients on dividend decreases appear to be more significant and larger in magnitude than those on dividend increases though not all of them are significant. Future research could possibly investigate what causes the difference between dividend increases and decreases, and whether dividend decreases are more valuable in assessing the firm's future earnings. Such further detailed research could provide additional insights about the information content of dividends.

References

Aharony, J. and Swary, I. (1980), *Quarterly Dividend and Earnings Annoucements and Stockholders' Returns: An Empirical Analysis*. The Journal of Finance, Vol. 35, No.1, pp. 1-12

Al-Yahyaee, K. H., Pham, T. M. and Walter, T.S. (2011), *The information content of cash dividend announcements in a unique environment*. Journal of Banking & Finance, Vol. 35, No.3, pp. 606-612

Alihud, Y. and Li, K. (2006), *The Declining Information Content of Dividend Announcements and the Effects of Institutional Holdings*. Journal of Financial and Quantitative Analysis, Vol. 41, No.03, pp. 637-660

Andres, C., Betzer, A. and Bongard, I. (2013), *The Information Content of Dividend Surprises: Evidence from Germany*. Journal of Business Finance & Accounting, Vol. 40, No. 5-6, pp. 620-645

Araujo, A., Moreira, H. and Tsuchida, M. (2011), *Do dividend changes signal future earnings?* Journal of Financial Intermediation, Vol. 20, No.1, pp. 117-134

Balachandran, B., Krishnamurti, C., and Theobald, M. (2012), *Dividend Reductions*, the *Timing of Dividend Payments and Information*. Journal of Corporate Finance, Vol. 18, No.5, pp. 1232-1247

Baker, H.K. (2009), *Dividends and Dividend Policy*. JohnWiley & Sons, Inc., Hoboken, New Jersey, pp. 35-177

Benartzi, S., Michaely, R. and Thaler, R. (1997), *Do changes in Dividend Signal the Future or the Past?* The Journal of Finance, Vol. 52, No.3, pp. 1007-1034

Bhattacharya, S. (1979), *Imperfect Information, Dividend Policy, and "The Bird in the Hand" Fallacy*. The Bell Journal of Economics, pp. 259-270

Braggion, F. and Moore, L. (2011), *Dividend Policies in an Unregulated Market: The London Stock Exchange*, 1895-1905.. Review of Financial Studies, hhr026

Brav, A. Graham, J. R., Harvey, C.R. and Michaely, R. (2005), *Payout Policy in the 21st Century*. Journal of Financial Economics, Vol. 77, No.3, pp. 483-527

Cao, Y., Myers, L. A. and Sougiannis, T. (2011), *Does Earnings Acceleration Convey Information*? Review of Accounting Studies, Vol. 16, No.4, pp. 812-842

Damodaran A. (2010), Applied Corporate Finance. John Wiley & Sons, pp.10.44

DeAngelo, H. and DeAngelo, L. (1990), *Dividend Policy and Financial Distress: An Empirical Investigation of Troubled NYSE Firms*. The Journal of Finance, Vol. 45, No.5, pp. 1415-1431

DeAngelo, H., DeAngelo, L. and Skinner, D.J. (1996), *Reversal of Fortune, Dividend Signaling and the Disappearance of Sustained Earnings Growth*. Journal of Financial Economics, Vol 40, pp. 341-371

DeAngelo, H., DeAngelo, L. and Stulz, R.M. (2006), *Dividend Policy and the Earned/Contributed Capital Mix: a Test of the Life-Cycle Theory*. Journal of Financial Economics, Vol. 81, No.2, pp. 227-254

Denis, D.J. and Osobov, I. (2008), Why Do Firms Pay Dividends? International Evidence on the Determinants of Dividend Policy. Journal of Financial Economics, Vol. 89, No. 1, pp. 62-82.

Easterbook, F. H. (1984), *Two Agency-cost Explanations of Dividends*. The American Economics Review, Vol. 74, No. 4, pp. 650-659

Fama, E. F. and French, K. R. (2001), *Disappearing Dividends: Changing Firm Characteristics or Lower Propensity Pay?* Journal of Financial Economics, Vol. 60, No.1, pp. 3-43

Freeman, R.N. and Tse, S. (1989), *The Multiperiod Information Content of Accounting Earnings: Confirmation and Contradiction of Previous Earnings Reports*. Journal of Accounting Research, Vol. 27, pp. 49-79

Grullon, G., Michaely, R., Benartzi, S. and Thaler, R.H. (2005), *Dividend Changes Do Not Signal Changes in Future Profitability*. Journal of Business, Vol.78, No.5, pp.1659-1682

Handjinicolaou, G. and Kalay, A. (1984), Wealth Distribution or Changes in Firm Value: An Analysis of Returns to Bondholders and Stockholders around Dividend Announcements. Journal of Financial Economics, Vol. 13, No. 1, pp. 35-63

Healy, P. M. and Palepu, K. G. (1988), *Earnings Information Conveyed by Dividend Initiations and Omissions*. Journal of Financial Economics, Vol. 21, No. 2, pp. 149-175

Jensen, G. R., Lundstrum, L. L. and Miller, R. E. (2010), *What do Dividend Reductions Signal?* Journal of Corporate Finance, Vol. 16, No. 5, pp. 736-747

Lang, L.H. and Litzenberger, R.H. (1989), *Dividend Announcements Cash Flow Signaling vs. Free Cash Flow Hypothesis?* Journal of Financial Economics, Vol 24, pp. 181-191

Lie, E. (2000), Excess Funds and Agency Problems: An Empirical Study of Incremental Cash disbursements. Review of Financial Studies, Vol.13, No.1, pp. 219-248

Nissim, D. and Ziv, A. (2001), *Dividend Changes and Future Profitability*. The Journal of Finance, vol. 56, No.6, pp.2111-2133

Koch, A.S. and Sun, A.X. (2004), *Dividend Changes and the Persistence of Past Earnings Changes*. The Journal of Finance, Vol. 59, No.5, pp. 2093-2116

Michaely, R., Thaler, R.H. and Womack, K.L. (1995), *Price Reactions to Dividend Initiations and Omissions: Overreactions or Drift?* The Journal of Finance, Vol50, No.2, pp.573-608

Miller, M. H. and Modigliani, F. (1961), *Dividend Policy, Growth, and the Valuation of Shares*. The Journal of Business, Vol. 34, No.4, pp. 411-433

Miller, M. H. and Rock, K. (1985), *Dividend Policy under Asymmetric Information*. The Journal of Finance, Vol. 40, No. 4, pp. 1031-1051

Pettit, R.R. (1972), *Dividend Announcements, Security Performance, and Capital Market Efficiency*. The Journal of Finance, Vol. 27, No.5, pp.993-1007

Ross, S.A. (1977), *The Determination of Financial Structure: The Incentive Signaling Approach.* The Bell Journal of Economics, pp. 23-40

Skinner, D.J. (2007), The Evolving Relation between Earnings, Dividends, and Stock Repurchases. Journal of Financial Economics, Vol.87, No.3, pp. 582-609

Tse, C. B. (2005), *Use Dividends to Signal or not: an Examination of the UK Dividend Payout Patterns*. Managerial Finance, Vol. 31, No.3, pp. 12-33

Watts, R. (1973), *The Information Content of Dividends*. The Journal of Business, Vol. 46, No.2, pp. 191-211

Yoon, P.S. and Starks, L.T. (1995), *Signaling, Investment Opportunities, and Dividend Announcements*. Review of Financial Studies, Vol.8, No.4, pp. 995-1018